Accelerator Status Report

Run Schedule: Jan. 3, 2011 through May 13, 2011
April 6, 2011
Injector    Energy 61.76 MeV
· The injector has been drifting, which is indicated by the transmission drop and interception increase. 

· (4/5) The Wien flipper was flipped from “right” to “left” for Hall C. 

· (4/5) EES took out the 0L07 spectrometer power supply to work on it. The power supply failed on April 2.

· (4/3) The laser spot was move to boost up current. After the spot move the injector was not stable. The transmission of beam keeps dropping which affects the beam quality. We have been doing a good job correcting the injector drift with auto-steer. We informed the injector experts of the situation.
· (4/2) 0L07 spectrometer power supply failed. The settings and trim card of corrector MAT0L07H were modified to give more steering power to compensate for the remanent field of 0L07 spectrometer magnet.
· (3/30) The limit of the PSS BCM in the injector was adjusted from 190uA to 357uA. Thursday swing shift the highest combined current out of the injector was 190uA (40uA to A and 150uA to C), first time ever for physics. 190uA was delivered for ~70 minutes. Hall A later requested a lower current.
· (3/24) The laser spot was moved to boost up beam current. After the spot move Reza set up high current to the inline dump. 
· (3/24) Troubleshot the injector 70MHz noise. The noise was reduced.

Accelerator   Linac Energy 549.00 MeV
· The accelerator runs well.
· (4/5) Beam studies: Wien flip change and some other test plans. When restoring beam after the beam studies Hall C beam was a stripe seen on the hall line viewers. We found the problem was caused by MBN3C04. We lowered the shunting current by 7%, which stabilized the beam.
· The accelerator has been stable.

· (3/29) Trim rack SL27B18 bulk ps fault. EES replaced it. (1.73 hrs.)
· (3/26) ARC6A tripped on DC overload fault. Beam was restored quickly (0.52 hrs.)

Hall A---Pass 4, E08 - 010
· No issues with Hall A beam.
· The beam was good. The hall has been taking low current (< 10uA). The hall will run higher current in a couple of days. 
Hall B ---Pass 3, Calibration of the Tagger
· (3/31) The hall finished the Tagger Calibration runs successfully. The hall will be offline until HDice is ready. When there is a power outage during the HDice experiment, the hall needs to go into the hall right away to save its target. 
· Set up 3-pass beam to Hall B Tagger dump. The hall needs about 6 days to take the calibration data. So far the hall has been doing well. HDice has been delayed for a couple of weeks. The beam will be terminated Thursday (3/31) until HDice is ready.  
Hall C--- Pass 1, QWeak 
· The beam (160uA) has been good. The thermocouple read backs have been stable. The Qtor trips off periodically.
· (4/2) Moller quad MQF3M01 misbehaved and EES resolve the issue.
· Hall C has been taking 150uA since swing shift last night and the hall is happy with beam. The shift plan standing order is updated regarding beam delivery to Hall C.

· (3/28) Monday evening there was a beam test after the vacuum work in the dump area. There are eight thermocouples attached to the flanges. The flanges are water cooled. The TCs appear to be working properly. We need to monitor the TCs while delivering beam to the hall (entry # 1576760). 
· (3/24) A beam test revealed that QTor steers beam to one o’clock direction when it is on, which is not by design.  This is most likely why there was a hot spot at one o’clock position on the flange (entry # 1576387). 

